

November 21, 2014

Dear Editor,

Please find enclosed the edited manuscript in Word format (file name: 14450-edited211114.doc).

Title: Therapeutic and prevention strategies against human enterovirus 71 infection.

Author: Chee Choy Kok

Name of Journal: World Journal of Virology

ESPS Manuscript NO: 14450

The manuscript has been improved according to the suggestions of reviewers:

1 Format has been updated

2 Revision has been made according to the suggestions of reviewers:

- (1) More subheadings have been included to make it easier for the reader to find specific topics.
- (2) Micro RNA inhibition of HEV71 is included in Page 20, Line 16.
- (3) Citation in Page 14, Line 11 has been corrected.
- (4) Tables 1-3 have been included to summarise and compare different therapeutic and prevention strategies.
- (5) The language has been edited.

3 References and typesetting were corrected

Thank you again for publishing my manuscript in the World Journal of Virology.

Sincerely yours

A handwritten signature in black ink, appearing to read 'C. Kok', with a stylized flourish extending from the bottom.

Chee Choy Kok, PhD

SEALS Microbiology

Level 4, Campus Centre

Prince of Wales Hospital

Randwick, 2031 NSW

Australia

Fax: +61-2-93829180

E-mail: Cheechoy.kok@sesiahs.health.nsw.gov.au